Author's response to reviews

Title: A randomized cross over trial of tolerability and compliance of a micronutrient supplement with low iron separated from calcium vs high iron combined with calcium in pregnant women [ISRCTN56071145]

Authors:

Eric Ahn (eric.ahn@utoronto.ca)
Nicholas Paireaudieu (npaireaudieu@hotmail.com)
Nicholas Paireaudieu Jr. (npaireaudieu@hotmail.com)
Yves Cerat (yvescerat@videotron.ca)
Bernard Couturier (bcouturier@videotron.ca)
Andre Fortier (jandrei.fortier@videotron.ca)
Eric Paradis (ericparadis65@hotmail.ca)
Gideon Koren (gkoren@sickkids.ca)

Version: 8 Date: 27 March 2006

Author's response to reviews:

March 27, 2006

Robin Stewart
BioMed Central
34-42 Middlesex House
Cleveland Street
London, W1T 4LB

Dear Robin:

In response to your e-mail message on March 27, 2006, enclosed please find a copy in which formatting changes have been made as requested, to comply with style.

Author list - The list of authors in the manuscript have been changed as they are in submission.

Title - Sentence case has been used in title.

Trial registration number - has been incorporated at the end of the title.

Author list - titles have been removed.

Affiliations - written in full.
Links to corresponding author have been made.

Underlining - all underlining has been removed from section headings.

Section headings - sentence case has been used.

Introduction - has been renamed to 'Background'.

References - cited using consecutive numbers in square brackets.

Patients and Methods - this section has been renamed 'Methods'.

Competing interests - has been changed to say 'All other authors declare they have no competing interests'.

Author's contribution - has been changed to Authors' contributions.

Tables - all tables have been moved to the end after references.

Thanks again for your help.

Sincerely,

Gideon Koren, MD, FABMT, FRCPC
Director, The Motherisk Program
Professor of Pediatrics, Pharmacology, Pharmacy,
Medicine and Medical Genetics
The University of Toronto
Senior Scientist, The Research Institute
The Hospital for Sick Children
The Ivey Chair in Molecular Toxicology
University of Western Ontario
Canada